Author’s response to reviews

Title: CLCN2-related leukoencephalopathy: A case report and review of the literature

Authors:

Zhuoxin Guo (guo_z_x@163.com)
Tingting Lu (lutingting Sysu@163.com)
Lisheng Peng (lisheng79peng@163.com)
Huanhuan Cheng (cheng544657716@163.com)
Fuhua Peng (pengfuhua93@163.com)
Jin Li (lijin8102@163.com)
Zhengqi Lu (lzq1828@protonmail.com)
Shaoqiong Chen (csq_q@yahoo.com)
Wei Qiu (weiqiu@protonmail.com)

Version: 2  Date: 30 Jun 2019

Author’s response to reviews:

Dear Prof. Brock Madsen:

Thanks very much for your kind work and consideration on publication of our paper. On behalf of my co-authors, we would like to express our great appreciation to you and the reviewers. We read your comments carefully and made revisions. Responses to comments are as follow:

1. Please indicate only one corresponding author on the title page. Please ensure that this is consistent with the indicated corresponding author in the submission data.

Answer: In the revised version, only one corresponding author was indicated (Title page section, line 19, page 1). Excepted on the title page, we did not indicated the corresponding author on any parts of the paper.
2. In accordance with BMC Submission guidelines please revise so that the Discussion and Conclusions sections are presented under one heading, “Discussion and Conclusions.”

Answer: We made corrections accordingly. Please find in the Discussion and Conclusions section, line 23, page 4, and line 10, page 10.

3. Please relocate the statement on funding currently found in the Availability of data and materials subsection to the Funding subsection.

Answer: We made corrections accordingly. Please find in the Funding section, line 9, page 11.

4. At this point please remove the CARE checklist and Cover letter from the file inventory as these are no longer needed at this stage of the editorial process.

Answer: We made corrections accordingly.

5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Answer: We made corrections accordingly.

We hope that the correction will meet with approval.

Once again, thank you very much for your comments and suggestions.